Author’s response to reviews

Title: TOPBP1 missense variant Arg309Cys and breast cancer in a German hospital-based case-control study

Authors:

Magda A Blaut (blaut.magda@googlemail.com)
Natalia V Bogdanova (bogdanova.natalia@mh-hannover.de)
Michael Bremer (bremer.michael@mh-hannover.de)
Johann H Karstens (karstens.jh@mh-hannover.de)
Peter Hillemanns (hillemanns.peter@mh-hannover.de)
Thilo Dörk (doerk.thilo@mh-hannover.de)

Version: 3 Date: 18 November 2010

Author’s response to reviews:

We have made the requested formatting changes throughout the revised manuscript. While doing this, we also spotted an error in the Figure 2 which shows mean ages but was designated with "median" ages. This has been corrected in the figure and in the corresponding text. We are looking forward to seeing the manuscript online soon.

Best regards from Hannover, Magda Blaut and Thilo Dörk.